Author's response to reviews

Title: Scaling up combined community-based HIV prevention interventions targeting truck drivers in Morocco: effectiveness on HIV testing and counseling

Authors:

Hakima Himmich (h.himmich@gmail.com)
Lahoucine Ouarsas (ouarsas.lahoucine@gmail.com)
Fatima Zahra Hajouji (fatimazahra.hajouji@gmail.com)
Caroline Lions (caroline.lions@inserm.fr)
Perrine Roux (perrine.roux@inserm.fr)
Patrizia Carrieri (pmcarrieri@aol.com)

Version: 6 Date: 2 April 2015

Author's response to reviews:

not needed,